MOSS 4 Test Readiness Review Minutes

March 30, 2004

ELEMENT ATTENDANCE:

ESDIS SSDO, ESDIS SOO, FDS, EDOS, EMOS, ASDC DAAC, TES SIPS, GES DAAC, MLS SIPS, HIRDLS SIPS, OMI SIPS, Averstar

NEXT TELECON:  April 5-8, Mon-Thurs at 4pm EST

Local participants Mon-Wed in S241; Thru in E103

DIAL-IN NUMBER:

USA Toll Free Number: 866-581-4760

USA Toll Number: +1-210-280-9695 (international)

PASSCODE: 574686
MEETING HIGHLIGHTS:

All TRR presentations can be obtained from the Aura Testing website: http://jupiter02.gsfc.nasa.gov/AuraTesting/ under MOSS 4 link. Only highlights, changes, or clarifications will be presented below.

SSDO:

Karen mentioned that SCS 49 has been added to the MOSS 4 data to handle TES SIPS requests during the dry run for additional GBAD data. Bill mentioned that we don’t know if it will work or not but we can try it.

Karen reviewed the MOSS 4 schedule highlighting:

· Formal Run –week of April 5th; Daily 4pm EST status meetings will be held (Mon-Thurs)

· Requirement status complete by Wednesday April 14th 

· Debrief – Monday April 19th (not our regularly scheduled meeting time)

· SORR dry run – April 27th (telecon)

· SORR formal run – May 11th 

FDS:

Dave Rohrbaugh presented the FDS slides. No FDS requirements are being verified during MOSS 4; they were verified in MOSS 3.  Dave confirmed with OMI SIPS that DOY 108 and 109 CH_ORBEVENT3 files were not needed. Dave asked that he and Clint Stone be added to the distribution list for FDS product requests. He also mentioned to contact Clint Stone first if FDS assistance is required during the test.

EDOS:

Ron reported that EDOS C4.6c (MOSS 4 configuration) has been operational for some time and was used during the dry run. EDOS has only one requirement to be verified during MOSS 4 (Req. 469 – secure distribution of RBDs to EOC). EDOS produced and distributed what was expected in accordance with the MOSS 4 timeline. EDOS plans to run all 49 SCSs from the local MOSS 4 test tape using internal replays. Ron mentioned that they would be monitoring for failed PDSs/EDSs deliveries to the DAACs. A fix for the problem is being delivered with EDOS C4.6d and will be tested through Operations and/or upcoming EGS tests. Karen asked Ron if he knew whether during the EGS test would products be delivered to us? Bill stated that he would check with George of MRTT. He also mentioned that Operations should be contacted first if there are problems. Ron and Nancy plans to cover 7am-7pm shifts.

EMOS:

Paul presented the slides via telecon. EMOS only has one requirement to verify for MOSS 4 (Req. 638 – secure receipt of RBDs from EDOS). They would produce and distribute GBAD carryout files (2 hr files on the even hour) to GES DAAC. No deviations, concerns or open items. Karen asked Paul to touch base with Ed Zebrowski and make sure that he understands the time intervals for the MOSS test. She also mentioned that during the dry run, at the beginning, the GES DAAC was not seeing c/o files but after Karen spoke with Ed, the files started to appear. The GES DAAC never saw the first two c/o files. Karen asked Paul if he could check with Ed on this also. Karen asked Marghi (OMI SIPS) if the directories for the OBEVENT files delivery had been straighten out with EMOS. Marghi stated that it was almost, she was asked by Curt Tilmes to have the files delivered to a different directory. Marghi now has to forward that information to Nelson of EMOS.

After the telecon, Paul sent an email stating that he talked with Ed Zebrowski at GSFC.  He (Ed) is aware of the upcoming MOSS 4 test and its timeline, and says that all issues from the Dry Run have been resolved.
ASDC DAAC:

Steve presented the slides for ASDC via telecon. He mentioned that ASDC is ready to verify their MOSS 4 requirements: secure ingest and distribution and operations performance (failed in MOSS 3). Bill commented on slide #4 that the number of EDSs from EDOS should be 26 vs. 33 (due to QL flags not set in 7 APIDs) and there were 13 L2 ESDTs vs. 14. Steve mentioned that they are still tracking the ASDC ECS distributions to JPL (MISR/TES). The ASDC to TES secure copy data rate remains a concern for full up operations. Steve also reported that this morning, ASDC successfully ingested L2s. After the telecon, the L1Bs ingested successfully. The OPS staff was added to the contact list, they had been involved in all the previous MOSS tests. Steve stated the DUE tool had been updated to send failed subscriptions via secure distribution.

TES SIPS:
Andy presented the slides via telecon. He pointed out that the SIPS software version R5.0 is new since MOSS 3. They will be verifying two MOSS 4 requirements (secure receipt/distribution of products) and will be retesting most of the previously tested requirements. Andy indicated that on slide #5 the plan column for TL1BN and TL1BL should be 64 (32 granules per day). He also mentioned that due to the errors in the simulated APID 1701, the data would have to be adjusted prior to use. 

GES DAAC:

Ed reported that the GES DAAC has 84 MOSS 4 requirements assigned to them. Secure ingest and distribution are in place and will be used for all Aura SIPS interfaces. He mentioned that NCR 39016 is currently installed and tested in TS2 and it passed the CCB today (3/30/04). It should be promoted to the OPS mode very soon. The GES DAAC plans no down time during the test. Ed reminded the group to call the Operation Controller first, if there is a problem.

MLS:
Audrey presented the slide via telecon. MLS will retest previous MOSS requirements as well as their MOSS 4 requirements. She mentioned that the requirements for distribution of PH, DAP, and QA will not be tested; pending investigation of usefulness and future support. They are investigating a password issue with MTMGW. The data date for the L2s and L3s would be 1996 due to the simulated inputs used to generate the products.

OMI SIPS:

Marghi presented the slides via telecon.  She mentioned that they would be getting a new delivery of their L1B software on June 7th. SIPS version 0.9.40 had just been installed and are going through regression testing. She stated that they would have one directory change; Curt Tilmes requested that the orbital events files be delivered to a separate directory. Marghi mentioned, on slide #6, that she needs to add the new internal product, OMORBDEF, to the files to be delivered to ODPS.

HIRDLS:

Ken presented the slides via telecon. He stated that HIRDLS are ready to test all pre-launch requirements for HIRDLS SIPS and HIRDLS NRTP at UCB; no additional requirements planned to be verified. As of March 30, 2004, the HIRDLS SIPS data management system is shut down while they recover from a security problem. Software is being installed on alternate hardware and HIRDLS expects to be fully operational well before MOSS 4. After the telecon, on April 1, Vince informed Karen that the HIRDLS SIPS is back in operation.

Averstar:

Scott reported for HIRDLS that he would re-test HIR1CAL product with the GES DAAC and order the product via EDG. For MLS, ee plans to order 5 L2s and all L3s. The DAP was deleted from the slide #5. For OMI, he plans to order L2s. See slides #6-7 for the specifics.

ODPS held in separate telecon March 30th at 10am EST.
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